

Trawl impacts on the relative status of biotic communities of seabed sedimentary habitats in 24 regions worldwide

Pitcher, C. Roland; Hiddink, Jan Geert; Jennings, Simon; Collie, Jeremy; Parma, Ana; Amoroso, Ricardo; Mazor, Tessa; Sciberras, Marija; McConnaughey, Robert A.; Rijnsdorp, Adriaan J.; Kaiser, Michel; Suuronen, Petri; Hilborn, Ray **Proceedings of the National Academy of Sciences of the USA**

DOI: 10.1073/pnas.2109449119

Published: 11/01/2022

Peer reviewed version

Cyswllt i'r cyhoeddiad / Link to publication

Dyfyniad o'r fersiwn a gyhoeddwyd / Citation for published version (APA): Pitcher, C. R., Hiddink, J. G., Jennings, S., Collie, J., Parma, A., Amoroso, R., Mazor, T., Sciberras, M., McConnaughey, R. A., Rijnsdorp, A. J., Kaiser, M., Suuronen, P., & Hilborn, R. (2022). Trawl impacts on the relative status of biotic communities of seabed sedimentary habitats in 24 regions worldwide. Proceedings of the National Academy of Sciences of the USA, *119*(2), Article e2109449119. https://doi.org/10.1073/pnas.2109449119

Hawliau Cyffredinol / General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

• Users may download and print one copy of any publication from the public portal for the purpose of private study or research.

- You may not further distribute the material or use it for any profit-making activity or commercial gain
- You may freely distribute the URL identifying the publication in the public portal ?

Take down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

1 Article Type: Research reports

- 2 Title: Trawl impacts on the relative status of biotic communities of seabed sedimentary habitats in 24
- 3 regions worldwide
- 4 Short title: Trawl impacts on regional sediment seabed status
- 5 Authors: C. Roland Pitcher^{1*}, Jan G. Hiddink², Simon Jennings³, Jeremy Collie⁴, Ana M. Parma⁵, Ricardo
- 6 Amoroso⁶, Tessa Mazor^{1,10}, Marija Sciberras^{2,11}, Robert A. McConnaughey⁷, Adriaan D. Rijnsdorp⁸, Michel
- 7 J. Kaiser^{2,11}, Petri Suuronen^{9,12}, Ray Hilborn⁶.

8 Affiliations:

- 9 ¹CSIRO Oceans and Atmosphere, Brisbane, Australia
- 10 ² School of Ocean Sciences, Bangor University, Menai Bridge, Wales, UK
- 11 ³ Centre for Environment, Fisheries and Aquaculture Science, Lowestoft, United Kingdom
- 12 ⁴ University of Rhode Island, Narragansett, Rhode Island, USA
- 13 ⁵ Center for the Study of Aquatic Systems, CENPAT-CONICET, Puerto Madryn Chubut, Argentina
- 14 ⁶ University of Washington, Seattle, WA, USA
- 15 ⁷ NOAA, Alaska Fisheries Science Center, Seattle, WA, USA
- 16 ⁸ WMR Wageningen UR, Ijmuiden, Netherlands
- 17 ⁹ UN Food & Agriculture Organisation, Rome, Italy
- 18 Present addresses:
- 19 ¹⁰ Currently: Department of Environment Land Water and Planning, Melbourne, Victoria, Australia
- 20 ¹¹Currently: Heriot-Watt University, Riccarton, Edinburgh, EH14 4AS, UK
- 21 ¹² Currently: Natural Resources Institute (Luke), Helsinki, Finland
- 22 *Corresponding author: roland.pitcher@csiro.au
- 23 Queensland Biosciences Precinct
- 24 306 Carmody Road, ST. LUCIA QLD 4067 Australia
- 25 Ph:+61(7)38335954, Fax:+61(7)38335501

- 26 Author contributions: RH, MK and SJ conceived the overall TBP Project. CRP conceived the RBS
- approach, conducted the assessments, and wrote the manuscript with input from all authors. RA
- 28 collated regional trawl-effort data with Kathryn Hughes, CRP, ADR, RAM, AMP. JGH and MS conducted
- 29 meta-analyses; extended and updated by CRP, JC and AMP.
- 30 Competing interests: The authors have received funds from a range of sources including governments,
- 31 foundations, NGOs and industry (see Acknowledgements) that have interest in conservation, sustainable
- 32 use, and effective fisheries management which may be perceived as a conflict of interest. However,
- 33 the authors declare that neither these or any other interests have directly or indirectly influenced the
- 34 objectivity of this paper, and the findings and conclusions in the paper are those of the authors alone,
- 35 independent of their organizations or funding sources.
- 36 Data availability: The underlying data used in this paper are available at
- 37 https://sites.uw.edu/trawlingbp/datasets/. All other data needed to repeat the analyses in the paper
- 38 are presented in the paper or the supplementary materials, or published in cited articles.
- 39 Classification: BIOLOGICAL SCIENCES: Sustainability Science; PHYSICAL SCIENCES: Environmental
- 40 Sciences.
- 41 Keywords: trawl impacts; trawl footprints; recovery; habitat sensitivity; spatial upscaling.
- 42 This PDF file includes:
- 43 Main Text
- 44 Figures 1 to 4
- 45 Tables none
- 46 Supplementary Figures S1 to S6
- 47 Supplementary Tables S1 to S5
- 48 Supplementary Dataset S1
- 49

50 Abstract (250 words max)

51 Bottom trawling is widespread globally and is known to impact seabed habitats. Risk assessments can 52 evaluate these impacts and support management decisions needed to ensure that trawling is 53 sustainable. However, trawl risks remain unquantified at large scales in most regions. We address these 54 issues by synthesizing evidence on impacts of different trawl-gear types, seabed recovery rates and 55 spatial distributions of trawling intensity in a quantitative indicator of biotic status (relative amount of 56 pre-trawling biota) for sedimentary habitats, where most bottom-trawling occurs, in 24 regions 57 worldwide. Regional average status relative to an untrawled state (=1) was high (>0.9) in 15 regions, but 58 <0.7 in three (European) regions and only 0.25 in the Adriatic Sea. Across all regions, 66% of seabed area 59 was unaffected by trawling (status=1), 1.5% was depleted (status=0), and 93% had status >0.8. Regional 60 seabed status was related to the total area swept annually by trawling, as a proportion of region area, 61 enabling preliminary predictions of regional status when only the total amount of trawling is known. 62 Seabed status was high (>0.95) in regions where catches of trawled fish stocks meet accepted benchmarks for sustainable exploitation, demonstrating that environmental benefits accrue from 63 64 effective fisheries management. Results highlight regions needing more effective management to 65 reduce exploitation and improve stock sustainability and seabed environmental status. This research advances seascape-scale understanding of trawl impacts in regions around the world, enables 66 quantitative assessment of sustainability risks, and facilitates implementation of an ecosystem approach 67 68 to trawl fisheries management globally.

69

70 Significance Statement (to be between 50 and 120 words)

71 We assessed the relative biotic status of seabed sedimentary habitats on continental shelves and slopes

- 72 in 24 regions worldwide where bottom trawling occurs. Seabed status differed greatly among regions
- 73 (from 0.25 to 0.999, relative to an untrawled state of 1). Fifteen regions had average status >0.9. Two-
- thirds of all seabed area assessed was unaffected by trawling, 93% had status >0.8, but 1.5% had
- 75 status=0. Total area swept by trawling was a strong driver of regional status, providing a relationship to
- 76 predict status from an estimate of the regional total amount of trawling. Seabed status was high in
- 77 regions where fisheries met benchmarks for sustainable exploitation, implying collateral environmental
- 78 benefits of effective fisheries management, and highlighting regions needing improved management.

79 MAIN TEXT

80 Introduction

- 81 Bottom-trawl fishing occurs worldwide and is the most extensive anthropogenic direct physical 82 disturbance to seabed habitats (1, 2). Towing trawl gears such as otter or beam trawls, or dredges along the seabed has a wide range of direct and indirect impacts on habitats, the broader ecosystem and the 83 services they provide (3, 4, 5, 6, 7, 8), and often is portrayed as a destructive fishing practice by some 84 85 environmental NGOs. However, bottom-trawl fisheries provide about a quarter of marine catch (9), 86 making substantial contributions to global food supply and livelihoods (10). Recognition of the wider 87 environmental consequences of fishing, including seabed impacts of trawling, has contributed to development of an "ecosystem approach to fisheries" (EAF, 11) that considers broader ecosystem 88 89 sustainability in balance with fishery production when managing fisheries. EAF principles are being 90 adopted widely into international and national policy commitments, fishery management plans, and 91 sustainable-seafood certifications (12). 92 Balancing fishery production and ecosystem sustainability, however, remains a globally challenging issue
- 93 - partly because the required indicators of ecosystem state often are unavailable or cost-prohibitive to 94 acquire at management scales. Consequently, a common approach has been to consider the risks of 95 fishing impacts using expert judgement and/or qualitative scoring approaches, which provide indicators 96 of relative risk (13, 14, 15). In contrast, quantitative methods provide continuous objective indicators of 97 ecosystem state, more useful for supporting management of fishery impacts under EAF (13, 14, 15, 16, 98 17). Quantitative methods require appropriate response indicators. In an evaluation of seven candidate 99 indicators (18), total seabed community abundance (biomass, and numbers of individuals) was the best 100 performing indicator of seabed state, meeting all nine criteria required for state indicators (19), and also relating directly to ecosystem functioning (15, 18). 101
- Implementation of EAF for bottom-trawl fisheries requires assessment of their impacts on the status of communities of seabed biota. We address this global challenge for EAF by quantifying a community abundance state indicator for seabed sedimentary (benthic) habitats on continental shelves and slopes in 24 large regions covering 7.92 million km² worldwide, accounting for 18.9% of the 0–1000m depth range (20) and 19.5% of all trawl landings (9) globally. We synthesize the required information regarding direct impacts of trawling and recovery rates (7, 8), distribution and intensity of bottom trawling (9) and mapped composition of seabed sediments (e.g. 21) in a quantitative model of the relative benthic status

- (RBS) of the seabed (14), recently recommended as the best performing of three quantitative indicators
 evaluated (15). We focussed on sedimentary habitats because they comprise the majority area of
 seabed (>99% of shelf and slope has a >1m layer of sediments (22)), most bottom trawling occurs on
 these habitats, and managers require these fisheries to be assessed and to be sustainable.
 The RBS model estimates the long-term abundance of biota relative to their untrawled abundance as a
 function (Eq. 1) of trawl-depletion rates (proportional reduction per trawl pass), recovery rates
 (maximum annual increase in proportional abundance), and trawling intensities (as swept-area ratio,
- 116 SAR). We quantified RBS in high-resolution grid cells (~1 km²) in each region after first updating the
- 117 previous series of meta-analyses used to estimate depletion and recovery rates (7) with additional data
- 118 (see Methods). We also estimate regional mean RBS as the average of grid-cell values, to provide a
- relative indicator of the overall state of regional seabeds.

120 Results

121 Parameter estimates

We estimated trawl-depletion rates of benthic communities in mud, sand and gravel habitats for each trawl-gear type by re-modelling the relationship (7) between depletion rates of biota (Table S1) and seabed penetration depths of different trawl gears, including some additional data and with different sediment habitat types as a factor (Table S2, Fig. S1). Average depletion rates ranged from 0.047 to 0.261 depending on gear and habitat (Table S3, Fig. S2). Otter trawls caused the lowest depletion, followed by beam trawls and towed dredges. Depletion rates were lower in sand than in gravel and

127 Tonowed by beam trawis and towed dredges. Depletion rates were lower in sand than in gravel at128 mud.

129 We estimated recovery rates of benthic communities in sedimentary habitats by re-analyzing the 130 relationship (7) of decreasing community relative abundance (as a combination of biomass and numbers 131 of epifauna and infauna) along a gradient of increasing trawling impact (Eq. 2), with some additional 132 data and including how the relationship depended on sediment types (Fig. S3A, Table S4). Recovery 133 rates were estimated (using the fitted model, Eq. 3) for an untrawled community so that RBS would 134 indicate the state of community compositions that existed on and in sediments prior to trawling; 135 including some slower growing, larger bodied, and longer-lived biota that are more sensitive to trawling. Average recovery rates ranged from 0.29 to 0.68 (lower confidence limits, CL=0.25–0.48) along a gravel 136 137 to mud gradient (Fig. S3B). Slower recovery with increasing gravel reflects the greater proportions of

138 longer-lived species found in more stable gravel habitats (23, 24). We used the mean and lower CL of 139 recovery estimates, representing a spectrum of more sensitive biota compositions, because of the 140 higher level of concern for sensitive biota. These rates correspond to a range of maximum longevities (see Figure 3 in 25) averaging 8 years in mud to 18y in gravel (and up to 11y-22y respectively for lower 141 142 CL recovery rates). Longevities in sand were intermediate (mean=10y, up to 14y for lower CL recovery). 143 Trawl SAR intensities of otter trawling, beam trawling and towed dredging, mapped for grid cells covering 24 regions for which adequate trawling data were available (9), differed greatly among cells (0-144 145 210y⁻¹, mean=0.42) as well as among regions (regional average SAR: range=0.005–11y⁻¹, mean=1.28, Table S5). SAR was aggregated among cells at larger scales, but at fine scales within small grid cells, most 146 147 trawling tends to be distributed approximately randomly (26, 27), producing a dynamic mosaic of 148 recently impacted, recovering, and undisturbed patches of seabed. Long-term, however, all patches are 149 expected to be trawled at the average SAR of each grid cell (27, 28). 150 We assigned trawl-depletion and recovery rates appropriate to the trawl gear and sediment type (Fig. 151 54) of each grid cell, mapped for each region using available sediment data (Table S5). We used these 152 rates with the grid-cell trawl SAR intensity values for each gear type in Eq. 1 to estimate cumulative

trawl impacts and RBS for each grid-cell and region (Table S5). Grid-cell RBS values range between 0–1; trawled cells have RBS<1, untrawled cells have RBS=1. The mean RBS estimate represents a linear relative index of benthic state for sedimentary habitats, corresponding to effects on biota that have an average sensitivity to trawling (among the range of sensitivities comprising typical communities in these habitats prior to trawling). The lower CL of RBS is indicative of status for biota types having upper CL sensitivity. Thus, RBS=0 does not imply that all biota are depleted; rather, that among the mix of biota present before trawling, those with average or greater sensitivity to trawling (the response indicated by

- 160 RBS herein) would be entirely depleted whereas more resilient types may remain.
- 161 Regional status

Regional RBS was lower in most European regions and higher in most non-European regions (range
0.247–0.999; Fig. 1 maps). Average RBS was <0.7 in three European regions: Adriatic Sea (0.25), West of
Iberia (0.60) and Skagerrak–Kattegat (0.63). These three regions also had the highest percentage of area
where RBS=0 (68%, 21%, 23% respectively; Fig. 1 pie-charts; Table S5). The European regions had <50%
untrawled area where RBS=1; lowest were the North Sea (11%), West of Iberia (16%), Adriatic Sea (17%)

and Irish Sea (18%). All non-European regions except Northern Benguela had average RBS≥0.95. Chile,
Australasia and Alaska had the highest average RBS and the largest untrawled areas (68%–93%).
In 17 regions, particularly in Europe, RBS of continental shelves was lower than that of slopes (Fig. 1).
Conversely, continental slopes had lower mean RBS than shelves for 7 regions outside of Europe,
particularly southern Africa and southeast Australia. These differences primarily reflect distributions of
trawling in shelf or slope areas (9).

173 Relationships between regional grid-cell RBS values (in decreasing order) and cumulative seabed area 174 (Fig. 2) show the proportion of seabed having any given status, provide more nuanced information 175 about grid-cell RBS than the regional average, and reflect spatial patterns of trawl impacts on different 176 habitats. The regional area at the point of departure from RBS=1 indicates the relative size of untrawled 177 versus trawled seabed and the area under the curves correspond to average RBS. The percentage of 178 regional area where RBS=0, indicates seabed depleted of pre-trawling biota that have average or higher 179 sensitivity to trawling. Steeper curves reflect areas where trawling is more concentrated. For example, 180 Northern and Southern Benguela have relatively short, steep upper curves compared with other regions 181 with similar average RBS because fisheries in these regions target a narrow depth band on the slope (cf. 182 Fig. 1). Longer, flatter upper curves reflect widespread low-intensity trawling. For example, the North 183 Sea has the smallest percentage of untrawled area but also the smallest percentage of depleted seabed 184 among European regions except West of Scotland. The Adriatic stands out with the lowest status and a 185 very steep and almost linear RBS curve, indicating that most trawlable ground in the Adriatic is heavily 186 trawled.

The uncertainty intervals of RBS curves (Fig. 2) arise from use of the mean and lower CL of estimated recovery rates, representing a spectrum of more sensitive sedimentary biota; hence they also indicate potential outcomes for a range of biota with recovery rates corresponding to maximum longevities of about 8–22 years. The majority of biota comprising seabed communities in sedimentary habitats are shorter-lived and more resilient (25); thus, the RBS uncertainty interval presented in Fig. 2 represents a more precautionary range of RBS outcomes.

RBS with uncertainty intervals can be used to frame risk assessments for trawling impacts. For example, if, as part of regional environmental objectives, an appropriate threshold for acceptable seabed status is defined, then RBS curves with uncertainty intervals can estimate the probability of status being above or below that threshold, thus informing the risk of environmental objectives not being achieved. As an 197 illustration, if an acceptable threshold is set at RBS>0.8 for >80% of regional area (Fig. 2), then in the case of the North Sea (region 6), the lower and upper 95% confidence limits for RBS at 80% of regional 198 199 area are 0.633 and 0.790 (mean=0.719), and the lower and upper confidence limits for the percentage of regional area having RBS=0.8 are 65.0% and 78.9% (mean=71.8%). Thus, the 95% confidence interval 200 201 for RBS is just below the illustrative 0.8-at-80%-area threshold and therefore there is >97.5% probability 202 that seabed status would not meet a threshold set at that level. Similarly, six other regions (1 to 5 and 7) 203 have >97.5% probability of not meeting this threshold; Northern Benguela has more than ~50% 204 probability and the Irish Sea almost 50% probability. Conversely, of the 24 regions, 15 would have <2.5% 205 probability of not meeting the example objective (i.e. the lower confidence limit for their RBS curves are 206 above the 0.8-at-80%-area threshold).

207 The differing status of gravel, sand and mud habitats (Fig. 3) reflects both their differing sensitivity to 208 trawling (Fig. S4), and the distribution of trawling. Within regions, sand habitats typically have higher 209 average RBS and smaller proportions of low RBS categories, due to their lower sensitivity than mud or 210 gravel. The biggest within-region differences among habitats are in the Irish Sea where mud status is 211 heavily reduced. Overall, eight regions have one or more habitats with average RBS<0.8 (14 habitats in 212 total, Fig. 3A); 21 habitats in 10 regions have <80% of area with RBS>0.8, whereas 51 habitats in 14 213 regions are above this threshold (Fig. 3B). Otter trawling is the most widespread and greatest 214 contributor to cumulative reductions of RBS (Fig. 3A), even though other gear types cause greater 215 depletion per trawl-pass. Beam trawling noticeably reduces RBS in the North Sea as does dredging in the 216 Irish Sea.

217 The total amount of trawling in a region is a strong driver of regional RBS (Fig. 4A). For regions that lack high-resolution spatial data for trawling and habitats, this relationship can be used to predict regional 218 219 RBS, with specifiable uncertainty, from information about the total amount of trawling and gear types 220 used. Such predictions under-estimate RBS for sand but over-estimate RBS for gravel and mud habitats 221 (Fig. 4A). They may also under-estimate RBS for tropical regions, where recovery rates might be faster, 222 but hence would be conservative. This approach can infer preliminary regional status and facilitate 223 prioritization of management needs, including in regions with higher levels of trawling effort such as 224 Southeast Asia. Further, Amoroso et al. (9) showed that where fishing exploitation is at or below that 225 needed to catch maximum sustainable yield (MSY: a widely accepted reference point for sustainable fisheries), regional SAR was ≤0.25. Here, where regional SAR is ≤0.25, average RBS is 95% likely to be 226 227 >0.91 (Fig. 4A). Average RBS is >0.91 for 15 of 24 regions. We also directly compare average regional RBS and an accepted indicator of the exploitation status of fish stocks (the ratio of fishing mortality *f* relative to the maximum sustainable fishing mortality f_{MSY} , see Methods) (Fig. 4B). There was a clear, though scattered, negative relationship between regional RBS and the ratio f/f_{MSY} of stocks. In regions where most stocks are managed sustainably (i.e. f/f_{MSY} <1) the average regional RBS is >0.95, suggesting that managing trawl fisheries for sustainable exploitation of fish stocks contributes substantially towards ensuring that seabed status is high.

234 While assessing the status of sedimentary habitats is critical to ensuring integrity of the majority area of 235 seabed ecosystems at the broadest scales, perhaps more concern surrounds rarer more sensitive 236 biogenic habitat types. However, suitable data for such habitats were not available for multiple regions. 237 Hence, although we took a precautionary approach by using recovery rates for untrawled seabed 238 community compositions and considering the lower confidence interval of RBS, we could not assess 239 highly sensitive habitat-forming biota types that can characterize Vulnerable Marine Ecosystems (VMEs: 240 29, 30). VME biota typically have distributions restricted to hard grounds, which may have low exposure 241 to trawling (31, 32), but they also have high trawl-depletion and slow recovery rates - hence 242 management seeks to prevent impacts on VMEs (29). RBS can be applied to VME biota, however, the 243 lack of data for their distribution, depletion and recovery must be addressed first — consequently there 244 are few cases where their regional status has been assessed (e.g. 31 and 33 using a dynamic Schaefer 245 model; 34 using RBS). Here, in lieu of assessing RBS for habitat-forming biota, we calculated the 246 percentage of each region where trawl SAR exceeded an estimated local-extinction threshold for highly 247 sensitive biota (at SAR>0.35, Fig. S5A). This ranges from 0.2% of seabed area in southern Chile to 82% in 248 the Adriatic Sea, and is >20% for 10 regions (all European regions and Northern Benguela). Areas for this 249 metric are very similar to areas of regional 'uniform' trawl footprints, as estimated by Amoroso et al. (9) 250 (Fig. S5A,B). Where they do differ (Fig. S5B), the threshold SAR needed to categorize cells as trawled that 251 yields an equivalent area as the uniform footprint remains indicative of extinction thresholds for highly 252 sensitive biota (Fig. S5C). Thus, area of the uniform footprint also corresponds closely to the area of 253 regions where highly sensitive biota cannot persist. Further, we also calculated that the percentage of 254 each region where trawl SAR was <0.07, allowing highly sensitive biota to maintain status >0.8, ranges 255 from 18% of seabed area in the Adriatic to 98% in southern Chile (Fig. S5A) and is >80% for 10 non-256 European regions.

257 Discussion and conclusions

We used a quantitative indicator of relative benthic status (RBS) to synthesize recent advances in understanding of trawling disturbance to the seabed and provide a seascape-scale assessment of cumulative trawl impacts on the relative biotic state of seabed sedimentary habitats, where most trawling occurs. Our results give insight into the sustainability of bottom trawling in 24 diverse regions around the world and provide comparisons for guiding regional management of environmental risks from trawling.

264 The RBS method is based on an established population dynamics model, widely applied in ecology and 265 for fisheries assessments, and has been recommended as the best performing method to assess trawling 266 impacts in sedimentary habitats (15). In our application to sedimentary habitats, RBS estimated status 267 based on relative abundance of combined seabed community biomass and numbers, which have been 268 shown to be the most suitable indicators of bottom-trawling impacts (18) as they respond strongly to 269 trawling, perform well against nine criteria for indicators (19), relate directly to ecosystem functioning 270 and health (15, 18, 35), and also account for the longevity composition of benthic communities, which 271 relates to structure and biodiversity (18). The depletion and recovery parameters used to estimate RBS 272 were sourced from meta-analyses of extensive seabed community data (7, 8) representative of the 273 composition of benthic invertebrate communities, including primarily biomass of epi-fauna, with some infauna and count data, as well as larger and longer-lived biota that may be more sensitive to trawling 274 275 (25). We estimated recovery rates applicable to pre-trawling community compositions, specifically 276 avoiding over-optimistic assessments compromised by small, fast growing, abundant species that may 277 dominate the more resilient biota associated with chronically trawled areas. Further, we took a 278 precautionary approach by considering the lower uncertainty intervals for recovery rates and RBS. 279 Our synthesis found that the biotic status of sedimentary habitat differs greatly among regions. In most 280 regions, some areas have low status, but large areas are little affected by trawling. Several regions, 281 primarily in Europe, had low habitat status relative to others, highlighting where management of 282 trawling could be prioritized to improve seabed environmental status. Twenty regions have average 283 status >0.8, a level that has been used as an impact limit threshold for VME habitats (36). These are first-284 order assessments but nevertheless provide important information that can be used to broadly compare 285 the extent of trawling effects on seabed status across multiple large-scale regions, different sedimentary 286 habitats, and different trawl gears.

287 The depletion and recovery parameters are derived from meta-analyses of multiple studies spanning 288 wide geographic areas, and are generalizable given that uncertainties are also characterized. These 289 uncertainties are substantive but are carried through to estimates of uncertainty in predicted status. 290 Nevertheless, these parameters can be refined to reduce uncertainty and to increase local specificity. 291 Regional assessments would benefit from analyses that were based on regionally specific depletion and 292 recovery rates, and definition, mapping of habitat types appropriate to their jurisdiction's sustainability 293 objectives (e.g. 31, 33, 34). Regionally determined parameters may also be able to account for additional 294 factors, such as potential temperature, productivity or depth effects on recovery, which were not 295 significant in the prior meta-analysis (7). The spatial extents of our regions were also relatively large and 296 likely to encompass substantive ecosystem heterogeneity. Ideally, eco-types could be defined 297 objectively at sub-regional scales to delimit the extent of status assessments (37). Our implementation 298 of RBS primarily considered direct impacts on benthic communities, rather than indirect impacts that 299 may affect other ecosystem components. Nevertheless, our estimation of recovery rates from larger-300 scale comparative studies of benthic communities along gradients of trawl-impact magnitude on actual 301 fishing grounds would account for indirect effects to the degree these impacted the sampled benthos. 302 Other indirect effects are possible, potentially including impacts on trophic relationships, nutrient 303 recycling or demersal fishes, among others (6). However, these indirect effects are more appropriately 304 assessed using other approaches (38); for example, food-web models (6); bycatch risk assessments (39); 305 and fishery stock assessments (38). These approaches together with RBS can provide more wholistic 306 assessments of ecosystem state.

Our assessment of relative status for sedimentary habitats achieved widest geographic coverage with
available data, but RBS is not limited to sediments or habitats. RBS can be used to assess status for
particular species or taxa including VMEs (34); communities based on taxonomic groups (35); and
benthos longevity classes (25). It is also possible to explicitly assess RBS for subsets of benthos
contributing different ecosystem functions (15, 35). In these cases, where continuous abundance
distributions are mapped, estimates of absolute status are possible (14, 31, 34, 35) cf. relative status as
herein for habitat classes.

- Unlike qualitative or categorical approaches that indicate relative risk (13, 14), RBS and other
 quantitative status indicators (e.g. 13, 15) enable risks of trawling on seabed status to be assessed
 against any defined sustainability thresholds, with transparency, objectivity and repeatability —
- 317 providing guidance to support management of trawling impacts (15, 16). Nevertheless, appropriate

318 thresholds for seabed habitats are undeveloped currently. It will require research, as well as broad 319 engagement between managers and society, to define thresholds that are consistent with sustainability 320 objectives and provide acceptable levels of precaution. RBS also enables evaluation of the effectiveness of alternative measures proposed to mitigate trawling risks (e.g. gear modifications or controls, effort 321 322 limitation, spatial management, 17). This would be achieved by simulating their implementation and 323 quantifying changes in predicted status. Such evaluations would facilitate management decisions 324 involving choice of measures needed to achieve environmental objectives (14, 17, 34) and trade-offs 325 with production (17).

326 We were not able to include all regions of the world where trawling occurs, due to either lack of high-327 resolution trawl-effort data, or because such data were not available for confidentiality reasons (9). 328 Where trawling data are confidential, regional authorities can apply RBS — and in regions that have only 329 fishery-scale trawl-effort data, regional SAR can be calculated from estimates of total area swept by 330 bottom trawling divided by total regional area and the strong relationship between regional SAR and 331 regional RBS enables preliminary estimates of status. Importantly, this relationship also indicates that if trawl target-species exploitation is managed sustainably, the reduced regional SAR will likely lead to high 332 333 seabed status. Hence, maximizing fisheries production within accepted sustainability limits and 334 sustaining the broader environment (EAF) are complementary goals and an objective balance between 335 them is demonstrably achievable.

336 Our approaches have important implications for regional environmental and fisheries management and

policy world-wide. They provide methods to address, and monitor progress towards, sustainability

338 objectives for trawl fisheries driven by international conventions, sustainable development goals (e.g.

339 UN SDG14: 'life below water'), national legislation (12), and sustainable-seafood certification

requirements for individual fisheries (e.g. 36). RBS provides a quantitative framework that can support

management decisions needed to balance fishery production with ecosystem sustainability and achievethe goals of EAF.

344 Methods

345 Study objectives and outline

- 346 We aimed to assess the status of sedimentary habitats because these habitat types comprise the majority of seabed area, contribute to integrity of seabed ecosystems at the broadest scales, are where 347 348 most bottom trawling occurs, and lack the quantitative status assessments that managers require. We 349 used the relative benthic status (RBS) model developed by Pitcher et al. (14) to estimate status, relative 350 to an untrawled state, of biotic communities that typify seabed sedimentary habitats exposed to chronic 351 trawling in 24 large regions worldwide where trawl footprints had been mapped by Amoroso et al. (9). 352 Trawling impacts on seabed habitats depend on the depletion caused by different gear-types, recovery 353 rates, distributions and exposure to trawling, thus defining the parameters and data required for quantifying the sustainability of trawling (40, 14). We estimated these parameters to implement RBS, 354 355 including trawl-induced depletion rates, and recovery rates, by updating a series of previous metaanalyses. Parameters were predicted for all trawl gear types (including, otter trawl, beam trawl, and 356 357 towed dredge) and for all combinations of percentage gravel, sand and mud that constitute sedimentary 358 habitats. 359 Trawl-gear depletion rates per trawl pass were derived from trawl-impact estimates for four gear types
- 360 provided by Hiddink et al. (7). We extended their existing meta-analysis to include sedimentary habitat types in addition to gear types, and some additional data. The extension was based on updating the 361 362 relationship between the penetration depth of gears into the sediments and the proportional rate of 363 depletion caused by each pass of the gear, where penetration depths were estimated for all 364 combinations of gear types and habitat types. Recovery rate parameters were derived by updating 365 another existing meta-analysis by Hiddink et al. (7). We extended that analysis, using a variation of their 366 model, and pooling data for both relative biomass and relative numbers as an overall measure of seabed 367 community relative abundance, after first including some additional data. All analyses were conducted 368 using the R Platform for Statistical Computing version 3.6.1 (41).

The wide availability of sediment mapping data enabled assessment of sedimentary habitats, which is where the majority of bottom trawling occurs. For other habitat types highly sensitive to trawling, lack of widely available distribution data precluded RBS assessment of status herein. Instead, we estimated the proportion of each region where highly sensitive, long-lived biota types could or could not persist due to chronic trawling.

374 Assessment model

375 We estimated the RBS of seabed habitats exposed to towed bottom-fishing gears, following the method 376 of Pitcher et al. (14). RBS is based on the Schaefer (42) production model, with an additional term to 377 describe the direct impacts of trawling on the seabed, consistent with previous seabed assessment approaches (28). The Schaefer model is commonly used in fishery assessments (e.g. 43), particularly in 378 379 data-poor situations where recently it has been demonstrated to be the least biased and most 380 frequently best performing for data-limited assessments globally (44), having excellent agreement with 381 results from more complex models (e.g. 45, 46). While this model is typically applied to single species, 382 management objectives and certification requirements also need to address seabed habitats and 383 communities in addition to species (36, 30). Pitcher et al. (14) reasoned that while habitats do comprise 384 many species with complex dynamics, previous studies have demonstrated that the aggregate properties of biotic communities in seabed habitats are relevant to characterizing trawling impacts (4, 385 386 5), and different sedimentary habitat types provide surrogates for their typical communities of 387 invertebrates, which form the basis of seabed ecosystems (47). Thus, the aggregate dynamics of seabed 388 communities in different habitats, integrated over benthos community composition and relevant time 389 frames and spatial scales, are parsimoniously described by the Schaefer model. Further, to enable 390 application to the typically data-limited circumstances of seabed assessment, Pitcher et al. (14) took the 391 simplifying approach that in habitats subject to chronic trawling, the long-term relative abundance of biota (B), as a fraction of carrying capacity (K) can be estimated by the equilibrium solution of the 392 393 Schaefer model:

B/K = 1 - F D/R where F < R/D, otherwise B/K = 0 Eq. 1

394	where B/K represents "relative benthic status" (RBS) of the seabed in the range 0–1, R is the
395	proportional recovery rate per year, which varies according to habitat, D is the depletion rate per trawl,
396	which depends on gear-type and habitat, and F is trawling intensity as swept-area ratio (SAR: the annual
397	total area swept by trawl gear within a given grid-cell of seabed, divided by the area of that grid-cell).
398	The ratio D/R represents sensitivity to trawling, the time interval between trawls (years) that would
399	cause local extinction of the biota (RBS=0) — and R/D is the corresponding critical annual trawl SAR
400	intensity F at which a given sensitivity will have RBS=0 (F_{crit}). Estimating RBS requires only parameters for
401	depletion and recovery rates, and distribution maps of trawling intensity and of habitat types. These
402	maps, and estimation of RBS, within an assessed region should be determined for grid cells of size $^{-1-3}$
403	$km^2 - a$ scale at which the distribution of most individual trawls has been shown to be random (26. 27.

404 48). At larger scales among cells of this size, patterns of trawling typically are aggregated and stable over 405 time (27, 49). Ellis *et al.* (28) distinguished two scales of depletion and recovery rates: *D* and *R* (as above) 406 are applicable at the grid-cell scale, whereas their analogues *d* and *r* are applicable at the scale of trawl-407 gears. If trawling is distributed randomly within grid cells then *D*=*d*; however, *R*<*r* and is related to *r* and 408 *d* through the equation *R*=*rd*/[-ln(1-*d*)] (28).

409 Trawl impact and depletion rates by gear type

410	Hiddink et al. (2) and Sciberras et al. (8) conducted meta-analyses of 46 experimental studies (n=152
411	records) of trawling impacts to estimate the proportional gear-scale depletion rate (d) of biota for each
412	pass of trawls of different gear types. They used a linear mixed-effects model (lme, R package nlme) to
413	analyse the change in biota abundance (pooled relative biomass and numbers of epifauna and infauna)
414	with time after experimental trawling, relative to the abundance before and/or in reference areas, as
415	log-response-ratio (InRR). Their results for the immediate loge trawl impact values (i: the intercept of
416	InRR at time 0) are directly related to depletion $(d=1-e^i)$ and represent the mean estimates for each gear
417	type across all habitat types (Table S1). Here, we have also estimated the standard errors on the natural
418	scale and 95% confidence limits (CLs) of the back-transformed <i>d</i> estimates for each gear type (Table S1).
419	Trawl penetration depth by gear and habitat types
420	Hiddink et al. (7) also showed that depletion rates of benthic-invertebrate communities were closely
421	related to the penetration depths (PD) of trawl gears into the sediments. Here, we re-analysed their
422	data (see Table S7 in 7) for PD of trawl-gear components by gear and habitat types, using the same log-
423	linear model but including the following data updates and additions: 1) records for Smith et al. (50) were
424	excluded as they reported PD of their sampling gear not trawl gear; 2) Freese et al. (51) reported PD for
425	the trawl ground-gear component, not whole gear, thus gear-width proportion was corrected from 1 to
426	0.25; 3) PD data were added from Rose et al. (52) for OT whole gear of 0.05 cm in mud habitat; and 4)
427	PD data were added from Depestele et al. (53) for BT whole gear of 4.1 cm in sand habitat. The final
428	dataset comprised 71 records from 48 studies. As per Hiddink <i>et al.</i> (7), we aggregated the model
429	estimates of mean PD for each gear component up to whole-gear estimates, weighted by the proportion

that each component comprises of the total gear-width — but whereas Hiddink *et al.* also aggregated
across habitats, we grouped by both gear and habitat to provide PD estimates on the natural scale for all

432 combinations of gear types and categorical sediment habitat types (Table S2, Fig. S1).

Field Code Changed

Formatted: Font: 11 pt, Not Bold
Formatted: Font: 11 pt, Not Bold

We propagated the uncertainty of the PD estimates for all gear components up to whole-gear estimates
by taking 2000 samples from the distributions of each gear-component mean, using standard deviations
as given by the standard errors of each mean reported by the fitted log(PD) model. The sampled
estimates of each gear-component PD were aggregated up to whole gear-by-habitat estimates, using
the same procedure as for the means, to provide 2000 estimates of PD for all combinations of gear and
habitat. The standard deviations of these estimates provide approximate standard errors for PD, and the
2.5% and 97.5% quantiles provide approximate 95% CIs for PD (Table S2, Fig. S1).

440 Trawl depletion vs penetration depth relationship

441 We estimated trawl depletion rates d for all combinations of gear types and sediment habitat types, by 442 re-fitting the same linear model of gear-mean depletion vs log of gear-mean PD relationship as Fig. 2 in 443 Hiddink et al. (7), but using the updated PD estimates for both gear and habitat (Table S2). Further, because the gear-mean d values estimated by Hiddink et al. (7) were from varying mixtures of habitat 444 445 types for each gear, for our estimates of the corresponding gear-mean PD across the three habitat types 446 for each of the four gear types (from Table S2) we calculated weighted mean PDs where the weights 447 were the frequency of studies by habitat for each gear type in the experimental studies meta-analysis 448 from which the four gear-mean d values were estimated (Table S1). This was done so that the estimates 449 of gear-mean PDs used to build the model (Fig. S2, grey dots) would correspond to the expected PDs of 450 the mixed habitat types represented in the meta-analysis that provided the d estimates.

451 The updated mean relationship between the depletion d of benthic community abundance and the 452 penetration depth PD of trawl gear was significant (Fig. S2, grey curve, R²=0.98), but with uncertainty. 453 The uncertainty of the model is indicated by the 95% CIs of the fit and by the prediction intervals (Fig. 454 S2, dashed grey lines and light grey shading). Additional uncertainties arise from the input data used to 455 build the model. These include the uncertainties for the gear-mean *d*-values (Table S1) and uncertainties 456 for gear-by-habitat PDs (Table S2, Fig. S1) plus additional uncertainty arising from calculating the 457 weighted-mean gear PD across habitats. The additional uncertainties are presented in Fig. S2, including: 458 the 95% CIs for each gear-mean d from the original experimental studies meta-analysis (vertical grey 459 lines; from Table S1); approximate estimates of the 95% CIs for gear-mean PDs propagated using the 460 procedure described below (horizontal grey lines); the 95% CIs for gear-by-habitat PDs (horizontal coloured lines; Table S2; Fig. S1); and approximate estimates of the 95% CIs for the predicted gear-by-461

habitat *d*-values (vertical coloured lines; <u>Table S3</u>), which were propagated from all sources of
uncertainty using the procedure described below.

464 The uncertainties for each gear-mean PD were propagated by sampling the 2000 estimates of each 465 habitat PD for each gear type (as described above) in proportion to the frequency of studies by habitat 466 for each gear type in the experimental studies meta-analysis (Table S1). The 2.5% and 97.5% quantiles of 467 these samples provide approximate 95% CIs for each gear-mean PD (horizontal grey lines, Fig. S2). These uncertainties and those for gear-mean *d*-values were propagated through the model by also sampling 468 469 and back-transforming 2000 estimates of each gear-mean impact *i* from the distributions of each mean 470 using the standard deviations given by the standard errors of each mean reported by the fitted InRR 471 model (Table S1). These two sets of 2000 sampled estimates for each gear-mean d and PD were used to 472 fit 2000 regressions, and each regression was used to predict d corresponding to each gear-by-habitat 473 mean PD (Table S2). The standard deviations of these predictions provide approximate standard errors 474 for each gear-by-habitat *d*, and the corresponding 2.5% and 97.5% quantiles provide approximate 95% 475 Cls (Table S3; vertical coloured lines, Fig. S2).

476 Recovery rates by habitat

477 In principle, experimental studies could also provide estimates of gear-scale recovery rates r. However, 478 these small-scale estimates may be overly-optimistic, especially for mobile fauna, due to short-distance 479 immigration from seabed adjacent to the experimental treatment. Grid-scale estimates of recovery are 480 preferable. Hiddink *et al.* (7) observed that Eq. 1 could be used to estimate recovery rates from large-481 scale comparative studies of trawling effects, which sampled the expected decrease in relative 482 abundance (B/K) of seabed communities on gradients of trawling intensity (F) on trawl grounds. The 483 slope of this relationship is D/R and if D is known from experimental studies then recovery R can be 484 estimated for the biotic community on trawl grounds. This approach assumes that the sampled benthos 485 populations are approximately in a balance between trawl impacts and recovery, under chronic 486 intensities of trawling, and that random processes and other departures are captured by the uncertainty 487 in the relationship. The assumptions are necessary given the scarcity of recovery information, and this novel approach represents a significant advance for estimating the grid-scale recovery rate R. Further, 488 489 they noted that trawling more rapidly depletes sensitive species and selects for species with faster life histories that are more resilient (40, 54), hence overall community R can be expected to increase with 490 the intensity F of chronic trawling — a response Hiddink et al. (7) found was approximated by a log-491

492 linear relationship between B/K and F: $\log_{10}(B/K) \sim bF$ — where the slope b of this relationship is a non-493 linear function of D, R and F.

494 Building on Hiddink et al. (7), we added to their data from 33 large-scale comparative studies (n=677 495 records) of trawl impacts on benthic invertebrate communities, and used an analogous meta-analysis, to 496 estimate recovery. Data for the meta-analyses were collated from published studies following a 497 systematic review protocol (55), which involved a high degree of control regarding the quality of studies 498 and to eliminate bias in selection of studies. We pooled data for relative biomass and relative numbers 499 as the response-ratio of overall relative abundance (B) since communities of benthic invertebrates 500 comprise a combination of both biomass and numbers of a wide range of species present. Further, 501 Hiddink et al. (18) found that community biomass and numbers were the most sensitive indicators of 502 the effects of trawling, and met all 9 criteria of indicator utility accepted in the literature (19). We first 503 updated the dataset with 27 additional records for three studies (2 numbers, 1 biomass) in gravel habitat from Collie et al. (56), and due the availability of improved data, we revised the trawling SAR 504 505 intensity (F) for sites sampled by two studies (56, 57) (see Data S1), and revised the sediment gravel, sand and mud fractions for these studies and seven others with information from Asch (58) and 506 507 Amoroso et al. (9) respectively. The final dataset totaled 711 records from 22 community biomass 508 studies and 14 numbers studies, and comprised 542 epifauna records and 169 infauna records, of which 539 were biomass records and 172 numbers. These data provided an extensive and representative 509 510 spectrum of the composition of benthic invertebrate communities, including larger and/or longer-lived 511 types of biota some of which would be sessile and sensitive types.

512 We fitted a variation of the Hiddink *et al.* (7) linear mixed-effects model to estimate how community

relative abundance decreased on a gradient of increasing trawling impact represented by the product *dF*

514 of depletion and trawl intensity:

$log_{10}(B/K) \sim b dF$ Eq. 2

515	This variation of the model was fitted so that recovery rates of different sedimentary habitats could be	
516	estimated without confounding by trawl gear types. For each study, d was calculated as a weighted	
517	mean of the habitat <i>d</i> values for the appropriate gear from <u>Table S3Table S3</u> , where the weights were	
518	the respective percentages of gravel, sand and mud fractions comprising each study's habitat. In a	
519	second model, we added covariates for the percentage gravel, sand and mud fractions of the habitat to	
520	estimate how the slope of the relationship changed with sediment composition (<u>Table S4</u> Table S4).	

Formatted: Font: 11 pt, Not Bold

Formatted: Font: 11 pt, Not Bold

521 Overall community relative abundance decreased with increasing trawling impact dF (Fig. S3A), with

522 each unit increase in dF leading to a mean decrease in abundance of 88.1%. The rate of decrease was

523 greater as the gravel content of the sediment increased relative to mud content.

524 The grid-scale recovery rate R was then estimated by equating Eq. 1 and Eq. 2 for B/K and solving for R, 525 giving:

$$R = dF/(1 - 10^{bdF})$$
 Eq. 3

526 and substituting the community slope b estimated by fitting model Eq. 2Eq. 2. To account for the 527 differing community compositions of different sedimentary habitats, the slope was varied with sediment 528 fractions according to the coefficients in <u>Table S4Table S4</u> (Fig. S3Fig. S3B). To account for the non-529 linearity of this relationship and to estimate recovery rates of pre-trawling community compositions, 530 which include larger/longer-lived sensitive biota, rather than higher R values associated with chronically 531 trawled community compositions of more resilient biota, dF in Eq. 3Eq. 3 was set close to zero (1×10⁻⁹). 532 The standard error of the slope b (Table S4Table S4) was used to estimate 95% confidence intervals of 533 the mean recovery rates along sediment gradients (Fig. S3Fig. S3B). 534 Trawl footprint and sedimentary habitat mapping 535 Mapping trawl footprints requires detailed information about trawling location and effort (e.g. hours of

536 trawling); however, in most countries, such information is confidential and not publicly available. 537 Amoroso et al. (9) approached management authorities in many regions to request access to data. 538 Ultimately, they were provided with high-resolution information for bottom-trawl fisheries in 31 539 regions, including continental shelves and slopes to 1000 m depth in Europe, North and South America, 540 Africa and Australasia. The information comprised satellite Vessel Monitoring System (VMS) and/or 541 vessel logbook data encompassing a period of several years (typically three years, 2008-2010). For each 542 fishing fleet, Amoroso et al. (9) also collated information about trawl gear types and sizes, and towing 543 speeds. From the product of trawling hours, gear-spread width and tow speed, they calculated swept 544 area ratio (SAR): the total area swept by bottom trawls each year within high-resolution grid-cell 545 locations (ca. 1 km² area each), divided by the area of those grid cells. For regions where collated data coverage was >70% of total trawl effort (24 of 31 regions), Amoroso et al. (9) used grid-cell SAR to 546 547 estimate trawl 'footprints', the area of seabed trawled one or more times in a given region and time 548 period. In addition, regional scale SAR can be calculated as the average annual regional total swept area Formatted: Font: Not Italic Formatted: Font: 11 pt, Not Bold Formatted: Font: 11 pt, Not Bold, English (United States) Formatted: Font: 11 pt, Not Bold, Not Italic, English (United States), Check spelling and grammar Formatted: Font: Italic Formatted: Font: 11 pt, Not Bold Formatted: Font: 11 pt, Not Bold, English (United States) Formatted: Font: 11 pt, Not Bold, Not Italic, English

(United States), Check spelling and grammar

549 divided by the total area of a region; these were also mapped by Amoroso et al. (9). The grid-cell scale 550 SAR is an area-standardized rate of trawling intensity (=F) that is an essential requirement for seabed 551 status assessment. To avoid under-estimating impacts, we scaled-up cell F by 100/coverage% for each 552 region and by gear type to approximate total trawl intensity, and re-calculated regional SARs and 553 footprints (Table S5). This scaling and re-calculation assumed the collated data are representative of the 554 spatial distribution of the total. In cases where the unavailable data may have a different distribution, 555 our assessment may slightly underestimate regional depletion, but less so than without scaling up. 556 Amoroso et al. (9) also collated data for seabed sediment composition from the comprehensive global 557 dbSEABED database of marine substrates (21) for most regions and from the MARS marine sediments 558 database (59) for Australian regions (Table S5). From these grain-size data (percentage of gravel, sand

559 and mud) they classified broad seabed habitat types to provide a consistent definition of habitat across all regions. Here we classified and mapped regional habitats as "gravel" if gravel%>30, else "sand" if 560 sand%>mud%, else "mud" to match the habitats for which depletion values were estimated by meta-561

analyses of experimental studies (7, 8).

563 Implementation of the status assessment

562

564 The predicted penetration depths (PD, Table S2), depletion values (d, Table S3) and predicted recovery 565 values (R, Fig. S3B) for otter trawls, beam trawls and towed dredges and for all possible combinations of percentage gravel, sand and mud that constitute sedimentary habitats are presented as ternary plots in 566 567 Fig. S4. For PD and d, these are weighted habitat means by gear, where the weights are the percentage 568 of gravel, sand and mud fractions of the sediment ternary distribution — this, in effect, provides continuous estimates of PD and *d*. Recovery *R* for sediment gradients was estimated using Eq. 3 and the 569 570 coefficients from Table S4 (Model 2). The ratio d/R indicates that gravel habitats are more sensitive to 571 trawling due to higher *d*-values and low *R*-values; sand habitats are less sensitive due to low *d*-values 572 and intermediate R-values; and mud habitats have intermediate sensitivity due to high d-values and high 573 *R*-values. All habitats are more sensitive to towed dredges and less sensitive to otter trawls. The ratio 574 R/d gives the critical threshold trawling intensity (F_{crit}) at which the estimated relative benthic status 575 (RBS) of the average untrawled community composition would be zero (Fig. S4). For the 24 regions defined by Amoroso *et al.* (9), we used these d and R values appropriate to the trawl 576 577 gear and sediment (Fig. S4), along with grid-cell SAR trawl intensity values F by gear type, substituted

578 into Eq. 1 to estimate RBS for each grid cell (expressed as a proportion of untrawled status between 0579 1). The estimated grid-cell RBS represents a mean estimate of the long-term relative abundance 580 (biomass and numbers combined) of the average compositions of biota typically present in different 581 sedimentary habitats prior to trawling, as sampled by the range of studies included in the meta-analyses 582 from which the parameters were estimated (i.e. primarily biomass of epifauna). We used the grid-cell 583 RBS values to assess the status of sedimentary habitats on the continental shelf and slope of each region 584 (Table S5). Trawl gear-types included otter trawls, beam trawls and towed dredges. Where more than 585 one gear type had fished a given cell, the cumulative RBS was estimated by summing the depletion 586 (Fd/R) due to the d and F values for each gear.

587 The region-wide status of sedimentary habitats, accounting for their different sensitivity and exposure 588 to trawling by different gear types, was summarized by mapping the regional average of grid-cell RBS 589 values, and by plotting the ordered distribution of grid-cell RBS values (high to low) against the cumulative proportion of regional area. To capture a range of uncertainty in estimating regional RBS, the 590 591 standard error of the slope b of Eq. 2 was used to estimate lower and upper 95% CLs for recovery R (Fig. 592 S3B). The regional RBS distributions were calculated using the mean and lower CL for R, to indicate the 593 range of status for average to more sensitive compositions of biota, including larger/longer-lived types, 594 typically present in sedimentary habitats prior to trawling. This was because of the higher level of 595 concern for sensitive biota, compared with more resilient biota that may be indicated by the upper CL 596 for R.

597 Relationship between RBS and sustainability of trawl fish stocks

598 For regions where stock assessment outputs were available for species targeted by bottom-trawl 599 fisheries, we examined the relationship between average regional RBS and a measure of the 600 sustainability of fishing on those stocks. A widely accepted indicator of the exploitation status of fish 601 stocks is the magnitude of fishing mortality (f) relative to the maximum sustainable fishing mortality 602 (f_{MSY}) at which fishery production is maximised over the long-term (maximum sustainable yield, MSY); $f_{\rm MSY}$ is considered a limit reference point and fishing exploitation rates are considered sustainable when 603 604 the ratio $f/f_{MSY} < 1$ (9). The mean f/f_{MSY} ratio for 2010–2012 was available for 87 individual trawled stocks in 12 of the 24 regions (Table S3 in Amoroso *et al.* (9), see <u>Table S5</u> for regional mean f/f_{MSY} ratios). We 605 606 plotted regional RBS against the ratio f/f_{MSY} and examined trends in the relationship (Fig. 4B).

607 Status of highly sensitive habitat types

608 Our primary regional RBS assessments were able to address status of sedimentary habitats due to the 609 wide availability of sediment data. We were not able to directly address the regional status of more 610 sensitive habitat-forming biota types, which can form Vulnerable Marine Ecosystems (VMEs, 29), due to the scarcity of large-scale distribution data for these types in multiple regions. VMEs are highly sensitive 611 612 to trawling because they have high depletion and slow recovery rates (29). In lieu of RBS, we estimated 613 the proportion of each region where trawling intensity SAR was too high for long-lived habitat-forming 614 biota to persist (if they had been present initially). Here, we define such biota as those that would have 615 RBS=0 at trawling SAR intensities F>0.35 (i.e. with $F_{crit}=R/d=0.35$, and the inverse: Sensitivity= $d/R=1/F_{crit}=2.86$), which corresponds to biota types with, for example, $d\approx 0.6$ and $R\approx 0.2$ — or 616 617 with $d\approx 0.3$ and $R\approx 0.1$ — or any other d/R ratio of about 2.86. Assuming the longevity relationship shown 618 in Figure 3 of Hiddink et al. (25), these examples would have maximum longevities of >25 yrs and >50 619 years respectively. This definition corresponds to the most trawl-sensitive of habitat-forming biota types assessed in previous case studies (e.g. >97th percentile of sensitivities for tropical taxa (31) and ~90th 620

- 621 percentile for temperate taxa (33)).
- 622 With this definition of highly sensitive biota, we calculated the percentage area of each region having
- 623 trawl SAR *F* exceeding 0.35 where such highly sensitive biota would have RBS=0 (Fig. S5A, bars). We also
- 624 calculated the percentage area of each region where sensitive biota, as defined, could persist with
- 625 status >0.8, which corresponds with where trawl SAR F was less than 0.07 (Fig. S5A, bar colours).

626 We also examined the comparability of the area of regions where *F*>0.35 with the regional trawl

- 627 footprints estimated by Amoroso et al. (9) (their "uniform" approach, i.e. sum of grid-cell areas A where
- 628 F>1 plus sum of A*F where F<1; see symbol '|' in Fig. S5A), which is indicative of a multi-year footprint.
- 629 We also plotted the difference in percentage of regional areas between these two calculations (Fig.
- 630 <u>55</u>B). Further, for each region, we also calculated the threshold *F* that would be needed to define cells as
- 631 "trawled" so that the total area of those grid cells corresponds to the area of the uniform footprint (<u>Fig.</u>
 632 <u>\$5</u>C).

633 Acknowledgements

This paper is an output of the "<u>Trawling Best Practices Project</u>". Chris Jenkins provided dbSeabed
 sediment data; regional fisheries agencies provided trawl-effort data; Michelle Bachman (NEFMC) and

- 636 David Keith (DFO) provided fishing intensity data for sampled sites on Georges Bank; comments from
- 637 Anthony Richardson and Eva Plaganyi improved the manuscript. Funding: the authors acknowledge their
- 638 organizations for salary support; the Walton Family Foundation, the David and Lucile Packard
- 639 Foundation, Food and Agriculture Organization of the United Nations, and several fishing industry
- 640 organizations for funding of regional workshops and travel.

641 References

642 643	1.	M. J. Kaiser, J. S. Collie, S. J. Hall, S. Jennings, I. R. Poiner, Modification of marine habitats by trawling activities: prognosis and solutions. <i>Fish Fish.</i> 3 , 114-136 (2002).
644 645 646 647 648 649	2.	 O. R. Eigaard, F. Bastardie, N. T. Hintzen, L. Buhl-Mortensen, P. Buhl-Mortensen, R. Catarino, G. E. Dinesen, J. Egekvist, H. O. Fock, K. Geitner, H. D. Gerritsen, M. M. González, P. Jonsson, S. Kavadas, P. Laffargue, M. Lundy, G. Gonzalez-Mirelis, J. R. Nielsen, N. Papadopoulou, P. E. Posen, J. Pulcinella, T. Russo, A. Sala, C. Silva, C. J. Smith, B. Vanelslander, A. D. Rijnsdorp, The footprint of bottom trawling in European waters: distribution, intensity and seabed integrity. <i>ICES J. Mar. Sci.</i> 74, 847-86 (2017).
650 651	3.	S. Jennings, M. J. Kaiser, The effects of fishing on marine ecosystems. <i>Adv. Mar. Biol.</i> 34 , 201-352 (1998).
652 653	4.	J. S. Collie, S. J. Hall, M. J. Kaiser, I. R. Poiner, A quantitative analysis of fishing impacts on shelf-sea benthos. <i>J. Anim. Ecol.</i> 69 , 785–798 (2000).
654 655	5.	M. J. Kaiser, K. R. Clarke, H. Hinz, M. C. V. Austen, P. J. Somerfield, I. Karakassis, Global analysis of response and recovery of benthic biota to fishing. <i>Mar. Ecol. Prog. Ser.</i> 311 , 1-14 (2006).
656 657	6.	J. S. Collie, J. G. Hiddink, T. van Kooten, A. D. Rijnsdorp, M. J. Kaiser, S. Jennings, R. Hilborn, Indirect effects of bottom fishing on the productivity of marine fish. <i>Fish & Fish.</i> 18 , 619–637 (2017).
658 659 660 661	7.	J. G. Hiddink, S. Jennings, M. Sciberras, C. L. Szostek, K. M. Hughes, N. Ellis, A. D. Rijnsdorp, R. A. McConnaughey, T. Mazor, R. Hilborn, J. S. Collie, C. R. Pitcher, R. O. Amoroso, A. M. Parma, P. Suuronen, M. J. Kaiser, Global analysis of depletion and recovery of seabed biota following bottom trawling disturbance. <i>Proc. Natl. Acad. Sci. U.S.A.</i> 114 , 8301-8306 (2017).
662 663 664 665	8.	 M. Sciberras, J. G. Hiddink, S. Jennings, C. L. Szostek, K. M. Hughes, B. Kneafsey, L. J. Clarke, N. Ellis, A. D. Rijnsdorp, R. A. McConnaughey, R. Hilborn, J. S. Collie, C. R. Pitcher, R. O. Amoroso, A. M. Parma, P. Suuronen, M. J. Kaiser, Response of benthic fauna to experimental bottom fishing: a global meta-analysis. <i>Fish Fish.</i> 19, 698-715 (2018).

666	9.	R. O. Amoroso, C. R. Pitcher, A. D. Rijnsdorp, R. A. McConnaughey, A. M. Parma, P. Suuronen, O. R.
667		Eigaard, F. Bastardie, N. T. Hintzen, F. A. Althaus, S. J. Baird, J. Black, L. Buhl-Mortensen, A.
668		Campbell, R. Catarino, J. S. Collie, J. H. Cowan Jr., D. Durholtz, N. Engstrom, T. P. Fairweather, H.
669		Fock, R. Ford, P. A. Gálvez, H. Gerritsen, J. A. González, J. G. Hiddink, K. M. Hughes, S. S.
670		Intelmann, C. Jenkins, P. Jonsson, P. Kainge, M. Kangas, J. Kathena, S. Kavadas, R. W. Leslie, S. G.
671		Lewis, M. Lundy, D. Makin, J. Martin, T. Mazor, G. G. Mirelis, S. J. Newman, N. Papadopoulou, P.
672		E. Posen, W. Rochester, T. Russo, A. Sala, J. M. Semmens, C. Silva, A. Tsolos, B. Vanelslander, C.
673		B. Wakefield, B. A. Wood, R. Hilborn, M. J. Kaiser, S. Jennings, Bottom trawl-fishing footprints on
674		the world's continental shelves. Proc. Natl. Acad. Sci. U.S.A. 115, E10275-E10282 (2018).
675 676	10.	FAO, <i>The State of World Fisheries and Aquaculture 2018</i> (Food and Agricultural Organisation, Rome, 2018).
677	11.	FAO. Fisheries management, 2. The Ecosystem Approach to Fisheries (FAO Tech. Guidelines for
678		Responsible Fisheries No. 4, Suppl. 2, Food and Agricultural Organisation, Rome, 2003).
679 680	12.	J. C. Rice, Evolution of international commitments for fisheries sustainability. <i>ICES J. Mar. Sci.</i> 71 , 157-165 (2014).
681 682	13.	C. R. Pitcher, Quantitative indicators of environmental sustainability risk for a tropical shelf trawl fishery. <i>Fish. Res.</i> 151 , 136-174 (2014).
683	14.	C. R. Pitcher, N. Ellis, S. Jennings, J. G. Hiddink, T. Mazor, M. J. Kaiser, M. I. Kangas, R. A.
684		McConnaughey, A. M. Parma, A. D. Rijnsdorp, P. Suuronen, J. S. Collie, R. O. Amoroso, K. M.
685		Hughes, R. Hilborn, Estimating the sustainability of towed fishing-gear impacts on seabed
686		habitats: a simple quantitative risk assessment method applicable to data-limited fisheries.
687		Methods Ecol. Evol. 8, 472-480 (2017).
688	15.	A. D. Rijnsdorp, J. G. Hiddink, P. D. van Denderen, N. T. Hintzen, O. R. Eigaard, S. Valanko, F.
689		Bastardie, S. G. Bolam, P. Boulcott, J. Egekvist, C. Garcia, G. van Hoey, P. Jonsson, P. Laffargue, J.
690		R. Nielsen, G. J. Piet, M. Sko ld, T. van Kooten, Different bottom trawl fisheries have a
691		differential impact on the status of the North Sea seafloor habitats. ICES J. Mar. Sci. 77, 1772-
692		1786 (2020).
693	16	A.D.M. Smith F.A. Fulton A.I. Hobday D.C. Smith P. Shoulder Scientific tools to support
694	20.	practical implementation of ecosystem-based fisheries management. <i>ICES J. Mar. Sci.</i> 64, 633–
695		639 (2007).
696	17	R A McConnaughey I G Hiddink S Jannings C R Ditcher M J Kaiser P Suuronan M Scibarras
697	17.	A. D. Riinsdorp, J. S. Collie, T. Mazor, R. O. Amoroso, A. M. Parma, R. Hilborn, Choosing best

698 699		practices for managing impacts of trawl fishing on seabed habitats and biota. <i>Fish Fish</i> . 21 , 319– 337 (2019).
700 701 702 703	18.	J. G. Hiddink, M. J. Kaiser, M. Sciberras, R. A. McConnaughey, T. Mazor, R. Hilborn, J. S. Collie, C. R. Pitcher, A. M. Parma, P. Suuronen, A. D. Rijnsdorp, S. Jennings, Selection of indicators for assessing and managing the impacts of bottom trawling on seabed habitats. <i>J. Appl. Ecol.</i> 57, 1199-1209 (2020).
704 705	19.	J. C. Rice, M. J. Rochet, A framework for selecting a suite of indicators for fisheries management. ICES J. Mar. Sci. 62 , 516-527 (2005).
706 707 708	20.	GEBCO Compilation Group, GEBCO global ocean gridded bathymetric data set. DOI: 10.5285/a29c5465-b138-234d-e053-6c86abc040b9 <u>https://www.gebco.net/data_and_products/gridded_bathymetry_data/</u> (2020).
709	21.	C. J. Jenkins, Building Offshore Soils Databases. Sea Techn. 38, 25-28 (1997).
710 711 712	22.	E. O. Straume, C. Gaina, S. Medvedev, K. Hochmuth, K. Gohl, J. M. Whittaker, R. Abdul Fattah, J. C. Doornenbal, J. R. Hopper, GlobSed: Updated total sediment thickness in the world's oceans. <i>Geochem. Geophys. Geosyst.</i> 20, 1756-1772. <u>https://ngdc.noaa.gov/mgg/sedthick/</u> (2019).
713 714 715	23.	 H. M. Tillin, J. G. Hiddink, M. J. Kaiser, S. Jennings, Chronic bottom trawling alters the functional composition of benthic invertebrate communities on a sea basin scale. <i>Mar. Ecol. Prog. Ser.</i> 318, 31-45 (2006).
716 717 718 719 720	24.	 S. G. Bolam C. Garcia, J. Eggleton, A. J. Kenny, L. Buhl-Mortensen, G. Gonzalez-Mirelis, T. van Kooten, G. Dinesen, J. Hansen, J. G. Hiddink, M. Sciberras, C. Smith, N. Papadopoulou, A. Gumus, G. Van Hoey, O. R. Eigaard, F. Bastardie, A. D. Rijnsdorp, Differences in biological traits composition of benthic assemblages between unimpacted habitats. <i>Mar. Environ. Res.</i> 126, 1-13 (2017).
721 722 723 724	25.	J. G. Hiddink, S. Jennings, M. Sciberras, S. Bolam, R. A. McConnaughey, T. Mazor, R. Hilborn, J. S. Collie, C. R. Pitcher, A. M. Parma, P. Suuronen, M. J. Kaiser, A. D. Rijnsdorp, Assessing bottom trawling impacts based on the longevity of benthic invertebrates. <i>J. Appl. Ecol.</i> 56,1075-1084 (2019).
725 726 727	26.	A. D., Rijnsdorp, A. M., Buys, F. Storbeck, E. G. Visser, Micro-scale distribution of beam trawl effort in the southern North Sea between 1993 and 1996 in relation to the trawling frequency of the sea bed and the impact on benthic organisms. <i>ICES J. Mar. Sci.</i> 55, 403-419 (1998).
728 729	27.	G. J. Piet, F. J. Quirijns, The importance of scale for fishing impact estimations. <i>Can. J. Fish. Aquat. Sci.</i> 66, 829-835 (2009)

730	28. N. Ellis, F. J. Pantus, C. R. Pitcher, Scaling up experimental trawl impact results to fishery
731	management scales—a modeling approach for a 'hot time'. Can. J. Fish. Aquat. Sci. 71, 1-14
732	(2014).
733	29. FAO, International Guidelines for the Management of Deep-sea Fisheries in the High Seas (Food and
734	Agricultural Organisation, Rome, 2009).
735	30. J. C. Rice, J. Lee, M. Tandstad, "Parallel initiatives: CBD's Ecologically or Biologically Significant Areas
736	(EBSAs) and FAO's Vulnerable Marine Ecosystems (VMEs) criteria and processes" in <i>Governance</i>
737	of Marine Fisheries and Biodiversity Conservation: Interaction and Coevolution, S. M. Garcia, J.
738	Rice A Charles Eds (Wiley New York 2015) nn 195-208
/30	nice, A. charles, Eds. (Whey, New York, 2015), pp. 155-200.
739	31. C. R. Pitcher, N. Ellis, W. Venables, T. J. Wassenberg, C. Y. Burridge, G. P. Smith, M. Browne, F. J.
740	Pantus, I. R. Poiner, P. J. Doherty, J. N. A. Hooper, N. Gribble, Effects of trawling on sessile
741	megabenthos in the Great Barrier Reef, and evaluation of the efficacy of management
742	strategies. ICES J. Mar. Sci. 73, i115-i126 (2016).
743	32. A. A. Rowden, O. F. Anderson, S. E. Georgian, D. A. Bowden, M. R. Clark, A. Pallentin, A. Miller, High-
744	resolution habitat suitability models for the conservation and management of vulnerable marine
745	ecosystems on the Louisville Seamount Chain. South Pacific Ocean. Front. Mar. Sci. 4. 335
746	(2017).
747	33. C. R. Pitcher, N. Ellis, F. Althaus, A. Williams, I. McLeod, "Predicting benthic impacts and recovery to
748	support biodiversity management in the South-east Marine Region" in Marine Biodiversity Hub,
749	National Environmental Research Program, Final report 2011–2015, N. J. Bax, P. Hedge, Eds.
750	(Report to Department of the Environment. Canberra, Australia, 2015;
751	http://nerpmarinebiodiversity2015.report/predicting-benthic-impacts-and-recovery-to-support-
752	biodiversity-management-in-the-south-east-marine-region/), pp. 24-25.
753	34. Australia and New Zealand, Cumulative Bottom Fishery Impact Assessment for Australian and New
754	Zealand Bottom Fisheries in the SPRFMO Convention Area, 2020. SPRFMO SC8-DW07-r1.
755	https://www.sprfmo.int/assets/2020-SC8/SC8-DW07-rev-1-Cumulative-Bottom-Fishery-Impact-
756	Assessment-for-Australia-and-New-Zealand.pdf (2020).
757	35. T. Mazor, C. R. Pitcher, W. Rochester, M. J. Kaiser, J. G. Hiddink, S. Jennings, R. Amoroso, R. A.
758	McConnaughey, A. D. Rijnsdorp, A. M. Parma, P. Suuronen, J. Collie, M. Sciberras, L. Atkinson, D.
759	Durholtz, J. Ellis, S. Bolam, M. Schratzberger, E. Couce, J. Eggleton, C. Garcia, P. Kainge, S. Paulus,
760	J. N. Kathena, D. Bowden, M. Gogina, D. van Denderen, A. Keller, B. Horness, R. Hilborn, Trawl
761	fishing impacts on the status of seabed fauna in diverse regions of the globe. Fish & Fish.
762	http://dx.doi.org/10.1111/faf.12506 (2020).

763 764	36.	MSC, MSC Fisheries Certification Requirements and Guidance Version 2.0 (Marine Stewardship Council, London, 2014).
765 766 767 768 769 770	37.	 C. R. Pitcher, W. Rochester, M. Dunning, T. Courtney, M. Broadhurst, C. Noell, J. Tanner, M. Kangas, S. Newman, J. Semmens, C. Rigby, T. Saunders, J. Martin, W. Lussier, <i>Putting potential</i> environmental risk of Australia's trawl fisheries in landscape perspective: exposure of seabed assemblages to trawling, and inclusion in closures and reserves — FRDC Project No 2016-039. CSIRO Oceans & Atmosphere, Brisbane, 71 pages. <u>http://www.frdc.com.au/Archived- Reports/FRDC%20Projects/2016-039-DLD.pdf</u> (2018).
771 772	38.	R. O. Amoroso, A. M. Parma, C. R. Pitcher, R. A. McConnaughey, S. Jennings, Comment on "Tracking the global footprint of fisheries" <i>Science</i> 361 , eaat6713 (2018).
773 774 775 776	39.	 A.J. Hobday, A.D.M. Smith, I. Stobutzki, C. Bulman, R. Daley, J. Dambacher, R. Deng, J. Dowdney, M. Fuller, D. Furlani, S.P. Griffiths, D. Johnson, R. Kenyon, I.A. Knuckey, S.D. Ling, C.R. Pitcher, K.J. Sainsbury, M. Sporcic, T. Smith, T. Walker, S. Wayte, H. Webb, A. Williams, B.S. Wise, S. Zhou, Ecological Risk Assessment for the Effects of Fishing. <i>Fish. Res.</i> 108,372-384 (2011).
777 778 779	40.	C. R. Pitcher, I. R. Poiner, B. J. Hill, C. Y. Burridge, The implications of the effects of trawling on sessile megazoobenthos on a tropical shelf in northeastern Australia. <i>ICES J. Mar. Sci.</i> 57, 1359- 1368 (2000).
780 781	41.	R Core Team. R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. URL <u>https://www.R-project.org/</u> . (2019).
782 783	42.	M. B. Schaefer, Some aspects of the dynamics of populations important to the management of commercial marine fisheries. <i>Bull. Inter-American Trop. Tuna Comm.</i> 1 , 27-56 (1954).
784 785 786	43.	C. Costello, D. Ovando, T. Clavelle, C. K. Strauss, R. Hilborn, M. C. Melnychuk, T. A. Branch, S. D. Gaines, C. S. Szuwalski, R. B. Cabral, D. N. Rader, A. Leland, Global fishery prospects under contrasting management regimes. <i>Proc. Natl. Acad. Sci.</i> USA 113 ,5125–5129 (2016).
787 788 789 790 791	44.	 A. A. Rosenberg, M. J. Fogarty, A. B. Cooper, M. Dickey-Collas, E. A. Fulton, N. L. Gutiérrez, K. J. W. Hyde, K. M. Kleisner, T. Kristiansen, C. Longo, C. Minte-Vera, C. Minto, I. Mosqueira, G. Chato Osio, D. Ovando, E. R. Selig, J. T. Thorson, Y. Ye, Developing new approaches to global stock status assessment and fishery production potential of the seas. FAO Fisheries and Aquaculture Circular No. 1086. Rome, FAO. 175 pp. <u>www.fao.org/3/i3491e/i3491e.pdf</u> (2014).
792 793	45.	S. Martell, R. Froese, A simple method for estimating MSY from catch and resilience. <i>Fish Fish</i> . 14 , 504–514 (2013).

- A. A. Rosenberg, K. M. Kleisner, J. Afflerbach, S. C. Anderson, M. Dickey-Collas, A. B. Cooper, M. J.
 Fogarty, E. A. Fulton, N. L. Gutiérrez, K. J. W. Hyde, E. Jardim, O.P. Jensen, T. Kristiansen, C.
 Longo, C. Minte-Vera, C. Minto, I. Mosqueira, G. Chato Osio, D. Ovando, E. R. Selig, J. T. Thorson,
- J. C. Walsh, Y. Ye, Applying a New Ensemble Approach to Estimating Stock Status of Marine
 Fisheries around the World. *Cons. Letters*, 12363 (2017).
- 47. T. Mazor, C. R. Pitcher, N. Ellis, W. Rochester, S. Jennings, J.G. Hiddink, R. A. McConnaughey, M. J.
 Kaiser, A. Parma, P. Suuronen, M. Kangas, R. Hilborn, Trawl exposure and protection of seabed
 fauna at large spatial scales. *Divers. Distrib.* 23, 1280-1291 (2017).
- 48. J. Lee, A. B. South, S. Jennings, Developing reliable, repeatable, and accessible methods to provide
 high-resolution estimates of fishing-effort distributions from vessel monitoring system (VMS)
 data. *ICES J. Mar. Sci.*, 67, 1260–1271 (2010).
- 49. N. T. Hintzen, G. Aarts, A. D. Rijnsdorp, Persistence in the fine-scale distribution and spatial
 aggregation of fishing. *ICES J. Mar. Sci.*, **76**, 1072–1082 (2018).
- 50. C. Smith, H. Rumohr, I. Karakassis, N. Papadopoulou, Analysing the impact of bottom trawls on
 sedimentary seabeds with sediment profile imagery. *J. Exp. Mar. Biol. Ecol.* 285, 479-496 (2003).
- 51. L. Freese, P. J. Auster, J. Heifetz, B. L. Wing, Effects of trawling on seafloor habitat and associated
 invertebrate taxa in the Gulf of Alaska. *Mar. Ecol. Prog. Ser.* 182, 119-126 (1999).
- 52. C. S. Rose, J. R. Gauvin, C. F. Hammond, Effective herding of flatfish by cables with minimal seafloor
 contact. *Fish. Bull.* 108, 136-144 (2010).
- 53. J. Depestele, K. Degrendele, M. Esmaeili, A. Ivanović, S. Kröger, F. G. O'Neill, R. Parker, H. Polet, M.
 Roche, L. R. Teal, B. Vanelslander, A. D. Rijnsdorp, Comparison of mechanical disturbance in soft
 sediments due to tickler-chain SumWing trawl versus electro-fitted PulseWing trawl. *ICES J. Mar. Sci.* 76, 312-329 (2018).
- 817 54. P. D., van Denderen, S. G., Bolam, J. G., Hiddink, S., Jennings, A., Kenny, A. D., Rijnsdorp, T. van
 818 Kooten, Similar effects of bottom trawling and natural disturbance on composition and function
 819 of benthic communities across habitats. *Mar. Ecol. Prog. Ser.* 541, 31-43 (2015).
- 55. K. M. Hughes, M. J. Kaiser, S. Jennings, R. A. McConnaughey, C. R. Pitcher, R. Hilborn, R. O.
 Amoroso, J. Collie, J. G. Hiddink, A. M. Parma, A. Rijnsdorp, Investigating the effects of mobile
 bottom fishing on benthic biota: a systematic review protocol. *Environ. Evid.* **3**, 23 (2014).
- 56. J. S. Collie, J. M. Hermsen, P. C. Valentine, F. P. Almeida, "Effects of fishing on gravel habitats:
 assessment and recovery of benthic megafauna on Georges Bank" in *Benthic habitats and the*

825 826		<i>effects of fishing</i> P. Barnes, J. Thomas, Eds. (American Fisheries Society, Bethesda, MD, 2005), Vol 41, pp. 325-343.
827 828	57.	B. E. Smith, J. S. Collie, N. L. Lengyel, Effects of chronic bottom fishing on the benthic epifauna and diets of demersal fishes on northern Georges Bank. <i>Mar. Ecol. Prog. Ser.</i> 472, 199-217 (2013).
829 830 831	58.	 R. G. Asch, "A photographic analysis of bottom fishing disturbance and microhabitat associations among the colonial epifauna of Georges Bank." thesis, GSO, Univ. Rhode Island, Narragansett, RI. (2006).
832 833 834	59.	V. Passlow, J. Rogis, A. Hancock, M. Hemer, K. Glenn, A. Habib, "Final Report, National Marine Sediments Database and Seafloor Characteristics Project" (Geoscience Australia, Record 2005/08, 2005).
835 836 837 838	60.	M. Bachman, P. Auster, J. Coakley, G. DePiper, K. Ford, J. Livermore, D. Packer, C. Quartararo, D. Stevenson, P. Valentine, A. Verkade, F. Restrepo, T. S. Smeltz, "Fishing Effects Model Northeast Region" (Final Report, New England Fishery Management Council, 2019); <u>https://www.nefmc.org/library/fishing-effects-model</u>

840 Figures

841 Fig.1



Fig. 1. Maps of regional average relative benthic status (RBS) for continental shelves (0-200 m depth)
 and slopes (200-1000 m) in 24 regions. Pie-charts show proportional area by region in six RBS category

intervals; the pie legend (centre) also indicates the average of category proportions across all regions.

846 Black boundaries indicate study regions (i.e. exclusive economic zones or fishery management

847 jurisdictions or large marine ecosystems). Region numbers and names follow Fig. 2.







852 percentage of regional area. Where RBS=1 at top/left indicates untrawled seabed; RBS=0 at

bottom/right indicates depleted seabed. The lower uncertainty interval is indicated by the band 853

854 between cell-mean RBS and the lower 95% confidence limit of cell-RBS. Dotted horizontal and vertical 855 lines at RBS=0.8 and 80% of region area indicate example thresholds. The regions legend is ordered by

856 regional average RBS.





regions, and reduction of RBS (=1-RBS) due to cumulative impacts of different trawl-gear types (stacked

grey bars); (B) percentage area of each regional habitat in six RBS category intervals. Vertical dotted line

indicates RBS=0.8 in A, and 80% of regional habitat area in B.

864 Fig.4



865

Fig. 4. Relationships for regional average RBS versus (A) regional SAR for all 24 regions, fitted relationship and prediction interval; with fitted relationships for sedimentary habitats, and continental shelves and slopes; vertical dotted line indicates SAR=0.25 (see text); and (B) stock exploitation as the ratio of fishing mortality (*f*) over f_{MSY} reference point for individual trawl fishery stocks in 12 regions for years 2010–2012 (*9*) and regional average f/f_{MSY} ; green vertical dotted line at f/f_{MSY} =1 indicates an

accepted sustainable upper limit on fishing rate; light-green shading emphasizes data for regions where most stocks are managed sustainably (f/f_{MSV} <1) and average RBS \geq 0.95; linear fit to all 87 stocks in 12

872 most stocks are managed sustainably (f/f_{MSY} <1) and average RBS \ge 0.95; linear fit to all 87 stocks in 12 873 regions: slope= -0.101, R²= 0.71, p<0.001; linear fit to 12 regional means: slope= -0.131, R²= 0.91,

874 p<0.001.

Supplementary Information for

Trawl impacts on the status of biotic communities of seabed sedimentary habitats in 24 regions worldwide

C. Roland Pitcher*, Jan G. Hiddink, Simon Jennings, Jeremy Collie, Ana M. Parma, Ricardo Amoroso, Tessa Mazor, Marija Sciberras, Robert A. McConnaughey, Adriaan D. Rijnsdorp, Michel J. Kaiser, Petri Suuronen, Ray Hilborn

*Corresponding author: roland.pitcher@csiro.au

This PDF file includes:

Supplementary Figures S1 to S6	2
Supplementary Tables S1 to S5	7
Supplementary Data S1. (Microsoft Excel file)	10

Supplementary Figures S1 to S6



Fig. S1. Estimated trawl gear penetration depth (PD) by categorical sediment habitat types and trawl gear type, with approximate propagated 95% confidence intervals (CI), OT = otter trawl, BT = beam trawl, TD = towed dredge, HD = hydraulic dredge



Fig. S2. Relationship between trawl gear penetration depth (PD) and proportional depletion *d* of benthic community abundance caused by a single pass of different trawl gears (means across habitat: grey points), with approximate propagated 95% confidence intervals (CI), and predictions for each sediment habitat type and trawl gear type: OT = otter trawl, **BT** = beam trawl, **TD** = towed dredge, **HD** = hydraulic dredge.



Fig. S3. (A) Relationship between overall community relative abundance and trawling impact (as the product *dF* of depletion *d* and trawl intensity *F*): the thick black line is the overall mean of the fixed effects, the darker grey shade indicates the 95% confidence interval for the mean of the fixed effects, the lighter grey shade indicates the 90% prediction probability interval of the random effects, which includes the different studies, gear types and habitat types. The dotted and dashed lines indicate how the mean relationship changes with increasing gravel:mud ratio of the sediments. (B) Relationship between predicted recovery rate (*R* yr⁻¹) and habitat gravel:mud ratio (ranging from 0% gravel:100% mud to 100% gravel:0% mud, with 0% sand), calculated from the slopes of the relationship in A at $dF = 1 \times 10^{-9}$).



Fig. S4. Ternary plots of predicted penetration depths (PD), depletion values (d), recovery values (R yr⁻¹) and critical trawl intensity (F_{crit} , where RBS=0) for each gear type and all combinations of sediment gravel, sand and mud composition.



Fig. S5. Results for highly sensitive biota types. (A) Bar plots of the percentage of each region's area with trawl SAR intensity *F*>0.35, where the relative benthic status (RBS) of highly sensitive biota types (those with *F*_{crit}≥0.35) would be zero. The colour scale indicates the percentage of each region's area with *F*<0.07 where status of highly sensitive biota would be >0.8. The symbol '|' indicates the regional trawl footprint, as percent of each region's total area, estimated using the 'uniform' method of Amoroso *et al.* (*9*). (B) Difference between percentage area of each region with *F*>0.35 and the uniform footprint. (C) Threshold trawl SAR intensity *F* required to categorize grid cells as trawled, the area of which is equivalent to the area of the uniform trawl footprint for each region.

Supplementary Tables S1 to S5

Table S1. Trawl impact values (*i*: as log-response-ratio, InRR) and log standard errors (In SE) estimated from themeta-analysis by Hiddink *et al.* (7), with corresponding depletion rates (*d*), back-transformed standard errors(SE *d*), and 95% confidence limits (CL), by trawl gear type: OT = otter trawl, BT = beam trawl, TD = toweddredge, HD = hydraulic dredge. Integers in columns under gravel, sand and mud indicate frequency of studiesby habitats and gear that contributed to the experimental studies meta-analysis.

C	Crowal	Cond	Mud	Impact		Depletion	сг <i>ч</i>	Lower	Upper
Gear	Gravei	Sand	iviud	InRR i	III SE	d d	SE U	95% CL	95% CL
OT	1	6	5	-0.058	0.111	0.057	0.105	-0.175	0.243
BT	0	4	0	-0.150	0.241	0.139	0.216	-0.388	0.466
TD	1	16	0	-0.225	0.070	0.202	0.056	0.084	0.305
HD	0	11	2	-0.532	0.104	0.412	0.061	0.278	0.522

 Table S2. Estimated mean penetration depth (PD, cm), approximate propagated standard errors

 (SE) and 95% confidence limits (CL), by categorical sediment habitat types and trawl gear type: OT

 = otter trawl, BT = beam trawl, TD = towed dredge, HD = hydraulic dredge.

Gear	Habitat	estimated PD	approx. SE	lower 95% CL	upper 95% CL
OT	Gravel	1.9	1.5	0.2	5.8
OT	Mud	2.0	1.3	0.3	5.3
OT	Sand	1.1	1.0	-0.1	3.7
BT	Gravel	3.0	1.4	1.0	6.2
BT	Mud	3.2	1.0	1.4	5.2
BT	Sand	1.9	0.6	0.8	3.2
TD	Gravel	5.2	3.0	1.7	13.7
TD	Mud	5.4	3.2	1.7	13.9
TD	Sand	3.5	1.9	1.2	8.3
HD	Gravel	17.8	10.3	6.1	44.9
HD	Mud	18.5	6.0	10.3	33.2
HD	Sand	12.6	3.9	7.3	21.9

Table S3. Estimated trawl depletion rates by gear type and sediment habitat type(predicted using relationship shown in Fig.S2), with approximate standard errors (SE)and 95% confidence limits (CL) propagated from the original experimental studies meta-analysis (Table S1) and penetration depth analysis (Table S2) as well as due to predictionuncertainty from the *d* vs PD relationship (Fig.S2). Trawl gear types: OT = otter trawl, BT= beam trawl, TD = towed dredge, HD = hydraulic dredge.

Gear	Habitat	predicted d	approx. SE	lower 95% CL	upper 95% CL
OT	Gravel	0.108	0.082	-0.073	0.241
ОТ	Mud	0.115	0.080	-0.060	0.245
OT	Sand	0.047	0.101	-0.176	0.208
BT	Gravel	0.174	0.065	0.028	0.280
BT	Mud	0.181	0.064	0.042	0.286
BT	Sand	0.113	0.080	-0.064	0.244
TD	Gravel	0.254	0.057	0.137	0.355
TD	Mud	0.261	0.057	0.145	0.364
TD	Sand	0.193	0.062	0.057	0.295

Table S4. Coefficients of two linear mixed models for the meta-analysis of data from comparative studies of relative changes in overall community abundance (pooled response-ratio of biomass and numbers) on a gradient of trawling impact (*dF*): *d*: depletion proportion per trawl; *F*: trawling intensity as swept-area ratio. Model 1) *dF* only as independent variable; Model 2) interaction of *dF* with gravel, sand and mud fractions (%+1) of sediments. DF: degrees of freedom.

Model	Co-variate	Slope (b)	Std.Error	DF	t–value	p–value
1	dF	-0.9256	0.1405	686	-6.5859	0.0000
2	dF:Gravel	-0.0145	0.0029	684	-4.9348	0.0000
2	dF:Sand	-0.0083	0.0026	684	-3.1393	0.0018
2	dF:Mud	-0.0061	0.0041	684	-1.4909	0.1365

Table S5. Summary of bottom trawling footprint, annual average over typically three years 2008-2010, by region, for depths of 0-1000 m. Numbers in the first column identify regions in the figures. Codes in parentheses for European regions indicate fishery management areas. Coverage (%) of total trawling activity in each region is estimated as per Amoroso *et al.* (9); here trawl activity data were up-scaled 100/coverage% to approximate total trawl effort. Regional sweptarea ratio (SAR) is the mean annual total area swept by trawl gears, after scaling-up, divided by the total area of the region to 1000 m depth. The uniform trawl footprint assumes that trawling is uniformly spread within grid cells and is indicative of a multi-year footprint. Sediment grain-size data (% gravel, sand and mud) were sourced from dbSEABED (21) or MARS (59) databases, from which gravel, sand and mud habitat types were classified. Relative benthic status results are the regional average of grid-cell mean RBS and the percentage (%) by area of each region with grid cell mean RBS>0.80, with RBS=0 and RBS=1.

	Region Name	Continent	Coverage %	Area (km²×10³)	Regional SAR	Footprint uniform %	Sediment source	Regional	%area of	%area of	%area of
#								mean	region	region	region
1	Adviction Soci (CECNA 2.1)	Furana	72	20 167	11 000	01.1		0.247	20.0	KB3=0	17.2
1	Auriatic Sea (GFCW 2.1)	Europe	72	39,107	LT:009	61.1		0.247	20.9	20.22	17.5
2	West of Iberia (ICES 9a)	Europe	100	40,303	2.335	00.1		0.590	40.4	20.85	10.1
3	Skagerrak and Kattegat (ICES 3a)	Europe	100	54,894	3.328	54.4	dbSEABED	0.633	55.1	22.60	26.7
4	Tyrrhenian Sea (GFCM 1.3)	Europe	82	137,924	2.787	51.9	dbSEABED	0.731	62.2	12.24	31.6
5	Western Baltic Sea (ICES 23-25)	Europe	72	87,070	1.282	38.9	dbSEABED	0.816	72.9	6.01	39.5
6	North Sea (ICES 6a,b,c)	Europe	86	586,108	1.215	52.1	dbSEABED	0.824	71.8	3.43	11.2
7	Aegean Sea (GFCM 3.1)	Europe	75	175,416	1.064	34.4	dbSEABED	0.834	74.4	5.06	47.6
8	Irish Sea (ICES 7a)	Europe	83	48,198	1.459	28.5	dbSEABED	0.836	80.6	9.10	17.9
9	North Benguela Current	Africa	95	203,002	1.018	28.0	dbSEABED	0.870	78.7	3.13	63.0
10	West of Scotland (ICES 6a)	Europe	81	160,640	0.506	24.2	dbSEABED	0.921	88.2	1.32	33.6
11	South Benguela Current	Africa	97	122,404	0.453	14.0	dbSEABED	0.949	91.7	0.72	70.1
12	Argentina	Americas	96	910,449	0.287	18.0	dbSEABED	0.966	96.0	0.13	54.7
13	East Agulhas Current	Africa	93	139,552	0.266	11.4	dbSEABED	0.967	95.2	0.44	61.8
14	Southeast Australian Shelf	Australasia	100	269,868	0.156	10.6	MARS	0.981	97.6	0.01	67.7
15	New Zealand	Australasia	90	1,052,723	0.118	9.2	dbSEABED	0.982	98.1	0.05	68.7
16	North California Current	Americas	100	119,327	0.107	10.0	dbSEABED	0.984	99.2	0.00	42.8
17	Northeast Australian Shelf	Australasia	100	529,357	0.129	6.7	MARS	0.985	97.9	0.17	78.6
18	East Bering Sea	Americas	97	797,969	0.073	6.4	dbSEABED	0.990	99.5	0.02	72.8
19	Aleutian Islands	Americas	97	94,721	0.026	1.8	dbSEABED	0.994	99.3	0.06	88.5
20	Gulf of Alaska	Americas	97	345,159	0.034	2.4	dbSEABED	0.994	99.4	0.02	87.7
21	Southwest Australian Shelf	Australasia	100	348,963	0.037	2.8	MARS	0.995	99.3	0.00	89.5
22	North Australian Shelf	Australasia	100	793,238	0.024	2.1	MARS	0.996	99.8	< 0.01	83.8
23	Northwest Australian Shelf	Australasia	100	679,604	0.024	1.7	MARS	0.997	99.7	0.01	93.0
24	South Chile	Americas	85	188,910	0.005	0.5	dbSEABED	0.999	99.9	0.00	92.6
		All regions		7,924,964	0.417	14.4		0.951	93.2	1.46	66.1

Supplementary Data S1. (Microsoft Excel file)

Data S1. Updated trawl-gradient studies data, including additional records from Collie *et al.* (*56*) (i.e. abundance data for US study sites, and biomass and abundance data for study sites in Canadian waters of Georges Bank), and revised otter trawl and scallop dredge swept-area ratio estimates for sites sampled by Collie *et al.* (*56*) and by Smith *et al.* (*57*) on Georges Bank. The updated otter trawl and scallop dredge SAR data for US study sites were provided by Michelle Bachman, New England Fishery Management Council (*60*), and updated otter trawl and scallop dredge effort data for Canadian study sites were provided by David Keith, Department of Fisheries & Oceans. These data replace those for StudyID's = 10, 42a & 42b in Hiddink *et al.* (*7*).

